Author’s response to reviews

Title: Pharmacokinetics of a single oral dose of vitamin D3 (70,000 IU) in pregnant and non-pregnant women.

Authors:

   Daniel E Roth (daniel.roth@sickkids.ca)
   Abdullah A Mahmud (mahmud@icddrb.org)
   Rubhana Raqib (rubhana@icddrb.org)
   Robert E Black (rblack@jhsph.edu)
   Abdulllah H Baqui (abaqui@jhsph.edu)

Version: 2 Date: 10 April 2012

Author’s response to reviews: see over
Dear Editor and Referees;

We mistakenly listed the overall aim of the study in the “Background” section of the abstract. We have since replaced it with what we feel is a more apt description of the context and purpose of this study.

Thank you for considering this manuscript. We look forward to your reply.

Sincerely, on behalf of the authors,

Daniel E. Roth MD PhD FRCP(C)
Associate Staff Paediatrician,
Division of Paediatric Medicine, The Hospital for Sick Children
Assistant Professor, Department of Paediatrics, University of Toronto
555 University Avenue, Toronto, ON M5G 1X8
Tel: 416-813-5795 Fax: 416-813-5663
Email: daniel.roth@sickkids.ca